

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL106636

Key Points:

- We analyze a new explicit spatial quantification of bias and uncertainty in the air-sea CO₂ exchange from observation-based SOM-FFN method
- We find variations in seasonal uncertainty with higher magnitude in boreal wintertime and larger uncertainty in less-observed regions
- Flux uncertainty is dominated by the exchange transfer velocity in polar regions and wind reanalysis estimates in the subtropics

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. Jersild, annika.jersild@mpimet.mpg.de

Citation:

Jersild, A., & Landschützer, P. (2024). A spatially explicit uncertainty analysis of the air-sea CO₂ flux from observations. *Geophysical Research Letters*, *51*, e2023GL106636. https://doi.org/10.1029/2023GL106636

Received 2 OCT 2023 Accepted 26 JAN 2024

Author Contributions:

Conceptualization: Peter Landschützer Formal analysis: Annika Jersild Funding acquisition: Peter Landschützer Investigation: Annika Jersild Methodology: Annika Jersild, Peter Landschützer Project administration: Peter Landschützer Supervision: Peter Landschützer Validation: Annika Jersild Visualization: Annika Jersild, Peter Landschützer Writing – original draft: Annika Jersild Writing – review & editing: Annika Jersild, Peter Landschützer

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

A Spatially Explicit Uncertainty Analysis of the Air-Sea CO₂ Flux From Observations

Annika Jersild¹ 🝺 and Peter Landschützer^{1,2} 🝺

¹Max Planck Institute for Meteorology, Hamburg, Germany, ²Flanders Marine Institute (VLIZ), Ostend, Belgium

Abstract In order to understand the oceans role as a global carbon sink, we must accurately quantify the amount of carbon exchanged at the air-sea interface. A widely used machine learning neural network product, the SOM-FFN, uses observations to reconstruct a monthly, $1^{\circ} \times 1^{\circ}$ global CO₂ flux estimate. However, uncertainties in neural network and interpolation techniques can be large, especially in seldom-sampled regions. Here, we present a three-dimensional (latitude, longitude, time) gridded product for our SOM-FFN observational data set consisting of uncertainties (pCO₂ mapping, transfer velocity, wind) and biases (pCO₂ mapping). We find that polar regions are dominated by uncertainty from gas exchange transfer velocity, with an average 48.7% contribution. In contrast, for subtropical regions, wind product choice contributes an average 50.0%. Regions with fewer observations correlate with higher uncertainty and biases, illustrating the importance of maintaining and expanding existing measurements.

Plain Language Summary The ocean plays an important role in regulating climate and the carbon cycle by absorbing and releasing carbon through the air-sea interface. In order to better understand these dynamics, we need to accurately quantify the amount of carbon exchanged between the ocean and atmosphere reservoirs, known as our air-sea carbon flux. Since the data can't be retrieved by satellites, it is challenging to get a global scale monthly product, so interpolation techniques such as neural networks are used. While these techniques have proven to provide robust observation-based estimates, uncertainties can be high, especially in regions where few observations are available. We calculate the uncertainty and bias created while using a two-step neural network machine learning method, the SOM-FFN. We find the sources of flux uncertainty vary regionally, with subtropical uncertainty dominated by choice of wind product but polar uncertainty influenced most by the coefficient chosen for the air-sea gas exchange transfer. Areas with fewer observations correlate with higher uncertainty and bias. This analysis provides important motivation for maintaining and increasing global ocean carbon observations, and is an important step toward closing the carbon budget through accurate quantification of the fluxes at the air-sea interface.

1. Introduction

As fossil fuel emissions continue to increase, carbon dioxide (CO₂) levels rise in our atmosphere, leading to changes in the air-sea carbon flux and alteration of the ocean carbon system (Landschützer et al., 2023; Sarmiento & Gruber, 2006). The ocean is a crucial component of global climate and climate change mitigation through its role in uptake of atmospheric CO₂. It has absorbed roughly 30% of anthropogenic carbon emissions, totaling about 2.9 ± 0.4 GtC yr⁻¹ in the last decade (Friedlingstein et al., 2023). The long-term ocean response to atmospheric pCO₂ varies depending on emission scenarios, with high emissions leading to growth of carbon sink and low emissions causing a decline as the ocean equilibrates with the atmosphere (Fay & McKinley, 2013; Gruber et al., 2023; Randerson et al., 2015; Ridge & McKinley, 2021). In order to make accurate predictions on the future of our global climate system and improve understanding of the oceans role as a climate regulator, we must quantify the amount of carbon exchanged at the air-sea interface, which is done using a combination of models and data-based estimates. In recent years though, we've seen a divergence between the two methods in the global carbon budget analyses (Friedlingstein et al., 2023; Hauck et al., 2020), which is a worrisome development.

In 2015, 196 countries signed the UNFCCC Paris Agreement, an international endeavor to mitigate climate change. With increased worldwide efforts to quantify the global stocktake, closing the carbon budget and constraining the uncertainty of our estimates is increasingly essential, as coastal regions could be a valuable carbon sink for many nations. However, ocean carbon is not currently a value that can be measured by satellites, so in order to span both spatial and temporal resolutions, other methods of calculation must be employed (Rödenbeck

19448007, 2024, 4, Dow

/10.1029/2023GL106636 by

et al., 2015). Observation-based products are useful tools to statistically interpolate the partial pressure of carbon dioxide (pCO₂) observations in order to resolve global-scale products (Fay et al., 2021; Rödenbeck et al., 2015). One product that closes the surface pCO₂ data gaps and is used in the Global Carbon Budget (Friedlingstein et al., 2023) is based on a two-step neural network approach, the self-organizing map feed-forward neural network (SOM-FFN) (Landschützer et al., 2013). The SOM-FFN reconstructs pCO₂ at a 1° × 1° resolution and a monthly timescale. From the resulting interpolated surface pCO₂ maps, the air-sea exchange is derived.

Despite the importance of understanding the air-sea carbon flux, observations are limited, and uncertainty in airsea transfer can be high (McKinley et al., 2023; Roobaert et al., 2019; Woolf et al., 2019). While the SOM-FFN estimates have proven to provide robust estimations on mean and seasonal timescales, uncertainties and biases in interpolation techniques can be large, especially in remote or seldom sampled ocean regions and on longer timescales (Gloege et al., 2021; Hauck, Nissen, et al., 2023). These uncertainties, consisting of mapping or extrapolation uncertainties as well as those associated with the kinetic transfer of gas across the air-sea surface, have the potential to impact our ability to balance regional and global carbon budgets. Despite this, they are seldom taken into account in synthesis efforts as they vary in time and space.

Here, we develop a three-dimensional (time, longitude, and latitude) gridded uncertainty product and analyze the regional and seasonal implications of the output. This allows for evaluation of high-uncertainty regions and identification of areas in the methods that may be further constrained. The open-source product provides tools to improve computations of air-sea CO_2 fluxes and uncertainties in support of regional carbon budgets. The identification and analysis of sources of uncertainty is important for improving ocean carbon flux products, and will contribute to a more accurately calculated global stocktake of ocean carbon.

2. Methods

2.1. The SOM-FFN Method

The SOM-FFN uses two stages to establish a relationship between independent predictor data and observed pCO_2 to output a monthly 1° × 1° resolution for global sea surface pCO_2 and the air-sea flux. While the method and its extension have been extensively described in (Landschützer et al., 2013, 2014), we provide a brief summary here highlighting key features. First, a neural network clustering algorithm (i.e., a self-organizing map) defines a set of biogeochemical provinces that share common relationships. These relationships are defined by clustering known driver data of pCO_2 (sea surface temperature (SST) (Huang et al., 2020), sea surface salinity (SSS) (Good et al., 2013), mixed layer depth (MLD) (Montégut et al., 2004)) and pCO_2 climatology (Takahashi et al., 2009) to create dynamical provinces designed to follow the seasonal CO_2 cycle in the ocean (Landschützer et al., 2013). Second, a feed-forward network (FFN) derives a non-linear, continuous relationship between pCO_2 measurements and driver data (SST, SSS, MLD) as well as chlorophyll-a (GlobColour, http://globcolour.info) and atmospheric CO_2 (Dlugokencky et al., 2021) for each biogeochemical province defined by step 1. This relationship is used to fill surface ocean pCO_2 gaps where no direct measurements exist. We calculate pCO_2 from Surface Ocean CO_2 Atlas (SOCAT) fugacity of CO_2 observations (Bakker et al., 2016).

We conduct our runs from 1982 to 2022, and calculate mean values for years 1990–2022 to constrain our analysis to temporal periods during which we see more consistency between different pCO_2 products (Fay et al., 2021; Rödenbeck et al., 2015).

2.2. Air-Sea Flux Calculation

Using the methods developed in (Landschützer et al., 2013, 2014, 2016), incorporating a standard quadratic bulk parameterization developed in (Garbe et al., 2014; Ho et al., 2006; Wanninkhof, 1992), we calculate the flux density (in molC $m^{-2} yr^{-1}$) from our reconstructed surface ocean pCO₂ using:

$$F_{CO_2} = k_w \cdot S_{CO_2} \cdot \left(1 - f_{ice}\right) \cdot \left(pCO_2 - pCO_2^{atm}\right) \tag{1}$$

where S_{CO_2} is the gas solubility of CO₂ in seawater, f_{ice} is fraction of sea ice cover (a dimensionless value between 0 and 1, (Rayner et al., 2003)), and pCO₂^{atm} is the pCO₂ at the air-sea interface. Atmospheric pCO₂ at 100% humidity was calculated following (Landschützer et al., 2013) using the atmospheric molar fraction of CO₂(xCO₂) data from the NOAA Marine Boundary Layer product (Dlugokencky et al., 2021), the NCEP sea

2 of 9







Figure 1. Reconstruction Bias and zonal averages averaged 1990–2022 (a) Total bias between reconstruction and model truth, shown in $molC/m^2/yr$. (b) Average zonal bias, over same time period and units. Values indicate magnitude of flux bias.

level pressure, and the water vapor correction following (Dickson et al., 2007). A positive flux is CO_2 outgassing from ocean into atmosphere, and a negative flux is ocean sink. Our baseline run adopts the mean gas transfer velocity following (Naegler, 2009), where we adjust the scaling factor to reach a global mean transfer velocity of 16.5 cmh⁻¹. The gas transfer velocity of CO_2 , k_w , can be defined using

$$k_w = a \cdot \langle U^2 \rangle \cdot (Sc/660)^{0.5} \tag{2}$$

Where Sc is the dimensionless Schmidt number, (a) represents the gas-transfer coefficient, and $\langle U^2 \rangle$ represents wind speed at 10m height (m/s).

2.3. Uncertainty Quantification and Data Incorporation

From these equations, we identify three major sources of bias and uncertainty highlighted in recent literature (Gloege et al., 2021; Hauck, Nissen, et al., 2023; Roobaert et al., 2018): (a) biases and uncertainty stemming from the mapping of unobserved ocean regions associated with the gradient in pCO_2 (mainly driven by surface ocean pCO_2), (b) uncertainty associated with *k* (the gas exchange transfer velocity), and (c) the uncertainty associated with choice of wind product used in Equation 2. Considering each source independent, we calculate overall uncertainty through:

$$\mu + \sigma = \mu + \sqrt{\sigma_{pCO_2}^2 + \sigma_k^2 + \sigma_{wind}^2}$$
(3)

Where μ represents bias or deviation from the true model pCO₂, and σ represents the random, unresolved uncertainty terms associated with the flux equation components. This statistical-based equation was developed following the linear model (e.g., Peterson et al., 2001), with further detail in the supplement.

To estimate σ_{pCO_2} we employ the approach by (Gloege et al., 2021), using the residuals from the MPIM-HAMOCC model (Ilyina et al., 2013), run in hindcast mode to represent realistic climate of the most recent decades. We subsample the MPIM-HAMOCC model along SOCAT measurement locations (Figure S1 in Supporting Information S1) and reconstruct the known model field with driver data from the model. Using the subsampled model data set, we run the SOM-FFN to interpolate a 1° × 1° surface ocean pCO₂, and repeat this 5 times, each realisation with a different, randomly chosen training and validation data set (see (Landschützer et al., 2013)). We then calculate the standard deviation of the difference between reconstructions and SOM-FFN output to quantify σ_{pCO_2} . We estimate μ by adopting the approach by (Hauck, Nissen, et al., 2023), using the mean pCO₂ deviation of the 5 reconstructions from the true model pCO₂. While this represents a small ensemble, we chose it to be comparable to the ensembles of the other σ terms. σ_k is quantified using the standard deviation across 4 commonly used quadratic parameterizations of the gas transfer coefficient ((Ho et al., 2006; Sweeney



Geophysical Research Letters



Figure 2. SOM-FFNv2022 air-sea CO_2 flux output averaged from 1990 to 2022. (a) Total air-sea flux density, shown in molC/m²/yr, generated using the standard observation-based SOM-FFN run. (b) Average uncertainty calculated from Equation 2, shown in molC/m²/yr. (c) Percentage uncertainty calculated for each point as flux uncertainty divided by the absolute value of the total flux.

et al., 2007; Takahashi et al., 2009; Wanninkhof, 2014)), and σ_{wind} is the standard deviation of the CO_2 flux across three wind products: ERA 5 Global Reanalysis (Hersbach et al., 2020), NCEP/DOE Reanalysis II (Kanamitsu et al., 2002), and NCEP-NCAR Reanalysis 1 (Kanlay et al., 1996). The supplement contains further specifications on data preparation and selection.

For analysis, regions are determined using the open-ocean biomes defined by Fay and McKinley (2014), which are classified using SST, spring/summer chlorophyll-a concentrations, ice fraction, and maximum MLD (Fay & McKinley, 2014). We combine their 17 biomes into 6 larger regions. While we focus on major uncertainty terms related to the extrapolation of sparse data and gas transfer formulation, there are terms we neglect here due to negligible influences on larger-scale fluxes. This includes the effect of scaling of the gas exchange in ice covered regions to the ice-free ocean area and the choice of SST product on gas solubility.

3. Results

Globally averaged, the μ term is negligibly small (-0.007 molC/m²/yr). This is somewhat contradicting toward results in (Hauck, Nissen, et al., 2023) that show despite compensating pCO₂ errors globally, errors in higher wind speed regions cause biases in the flux as well. We explain this difference through both the use of a different model and potentially the use of a single realisation compared to the mean of several runs used here. This difference is



Geophysical Research Letters



Figure 3. Seasonal Uncertainty, broken down by region. Note that Pacific equatorial region includes both east and west Pacific. (a–f) Shows climatology of CO_2 flux (molC/m²/yr) broken down by regions, defined based on (Fay & McKinley, 2014). Time series shows seasonal average with shaded uncertainty of the mean. (g) Shows mean July uncertainty averaged across 1990–2022. (h) Shows mean January uncertainty averaged across 1990–2022.

also reflected in the σ_{pCO2} uncertainty below, which shows that single realizations can differ substantially from each other. At regional scales, however, we observe larger variation ±0.5 molC/m²/yr (Figure 1a), which are comparable to errors observed in Gloege et al., 2021, although estimated from different model types and scenarios (i.e., large ensembles vs. hindcast simulations). This gives us confidence we can faithfully pinpoint biases in our reconstruction. Figure 1b further shows zonal averages, demonstrating higher bias in regions with a lower density of observations (observation density Figure S1 in Supporting Information S1). While our reconstruction bias doesn't directly impact the uncertainty quantification globally due to compensating errors, considering the influence of data sparsity on the flux calculations is critical for understanding the variability in our calculations, particularly on regional scales.

Figure 2 shows maps of the 2022 SOM-FFN data product, which is available through the NOAA National Centers for Environmental Information (Jersild et al., 2023). We see (a) air-sea flux as well as (b) flux uncertainty and (c) percentage of mean flux, all averaged from 1990 to 2022. We observe higher flux uncertainty in polar regions and the Pacific equatorial, but find the Southern Ocean and equatorial coastal regions dominate in proportional flux uncertainty. Part of this is due to lower regional fluxes, making small amounts of uncertainty more significant. In the Southern Ocean, where we have almost 45% uncertainty throughout the region, we also have a lower number of observations contributing to increased variance in the SOM-FFN method, which may contribute to higher uncertainty. We also see higher uncertainty in polar regions and areas (seasonally) covered in sea-ice, which provides important motivation to continue to increase our observations of the harder-to-observe but climatically pivotal regions.

Many of the regional errors balance each other, leading to overall smaller uncertainty when looking at the globally integrated air-sea CO_2 flux. The global average uncertainty from 2012 to 2022 is ± 0.31 GtC/yr, which is lower but comparable to Global Carbon Budget 2023 (Friedlingstein et al., 2023) estimation of ± 0.4 GtC/yr. That being





Figure 4. Breakdown of uncertainty sources, compared seasonally. (a) July uncertainty sources (averaged 1990–2022) color indicating dominating source of uncertainty (k, pCO₂, and wind). (b) Percentage each source contributes to total uncertainty, averaged latitudinally. (c, d) Same as (a, b) but for January.

said, while we do investigate uncertainty in the contemporary fluxes, we employ different methods and do not account for uncertainties in river fluxes or coastal regions as they do in the Global Carbon Budget. Temporally, the range of average uncertainty spans from a minimum of ± 0.21 GtC/yr in 1997 and a maximum of ± 0.42 GtC/yr in 2021, with seasonal variation.

Seasonally, we find average increases in uncertainty in January in both hemispheres (Figure 3), with a global average increasing from ± 0.25 GtC/yr in July to ± 0.37 GtC/yr in January. Although opposite seasons, we have different sources of uncertainty dominating these average values (Figure 4). We further break down into six ocean regions to compare average seasonal uncertainty at regional scales (Figure 3, Table 1). Uncertainty is greater in high latitudes for both hemispheres, with year-round averages of ± 0.35 GtC/yr for the Southern Ocean and ± 0.51 GtC/yr for the northern hemisphere high latitudes (see Table 1), both averaging an uncertainty around 40% of total flux.

Using our calculation from Equation 3, we identify the contributions of each source of uncertainty as k-dominated (driven by the k_w gas transfer velocity coefficient), pCO₂-dominated, or wind-dominated. Figure 4 shows the regional and seasonal breakdown of each of these drivers on overall uncertainty. We find that regardless of season, uncertainty in polar regions (averaged poleward of 60° north and south) is driven by the uncertainty associated with k_w , contributing 48.7% to total uncertainty (Table 2). In contrast, the Subtropical region centered around the 30° latitude has wind-driven uncertainty contributing about 50% to overall uncertainty (Figure 4, Table 2) However, smaller regions do experience seasonal variation. The North Atlantic, which sees a strong increase in uncertainty during wintertime, experiences a transition seasonally from mixed sources to wind-dominated (Figure 4). Increased wintertime wind speed drives this seasonal uncertainty and is a strong contribution to increased standard deviation in flux in this region.

Table 1

Seasonal Comparison of Flux (molC/m²/yr) and Uncertainty (%), Broken Down by Regions Northern Hemisphere High Latitudes (NHHL), Northern Hemisphere Subtropics (NHST), Atlantic Equatorial (Atl. Eq.), Pacific Equatorial (Pac. Eq.), Southern Hemisphere Subtropics (SHST), and Southern Ocean (S.O.)

	January flux	% uncertainty Jan	July flux	% uncertainty July
NHHL	-2.11 ± 0.79	37%	-0.93 ± 0.24	26%
NHST	-1.15 ± 0.21	18%	0.31 ± 0.12	38%
Atl. Eq.	0.49 ± 0.22	45%	0.56 ± 0.19	34%
Pac. Eq.	1.14 ± 0.27	24%	1.34 ± 0.29	22%
SHST	0.1 ± 0.15	150%	-0.82 ± 0.24	29%
S.O.	-1.23 ± 0.36	29%	-0.64 ± 0.34	53%

Note. January and July chosen to show 6 months out of phase across our decadal calculations.

4. Conclusions

Table 2

Here, we provide an explicit spatial quantification of the uncertainty in the air-sea CO_2 exchange from a commonly used observation-based method and thus provide a step forward in our ability to (a) understand modeldata differences, (b) provide a realistic outlook regarding global and regional stocktake efforts and (c) provide explicit spatial and temporal information where efforts need to be undertaken to reduce present day uncertainties. While previous studies have investigated uncertainties in the air-sea CO_2 exchange (Gloege et al., 2021; Hauck, Nissen, et al., 2023; Roobaert et al., 2018) we, for the first time, combine these well established terms to provide a view of current biases and uncertainties in the air-sea CO_2 exchange for an observational product on both a temporal and spatial scale. This new addition provides great opportunities for future research focusing on both temporal or regional-scale analysis.

We find that effective biases (μ) that result from our single model reconstruction by interpolation of sparse observations are small at global scale when the mean of multiple realizations are investigated, but can be substantial on regional scales. Global flux uncertainty is strongly seasonally dependent, with higher average values in January than in July. However, these regionally averaged patterns are driven by different sources of uncertainty, as we see increased austral summertime uncertainty in the gas-exchange-dominated Southern Ocean and increased boreal wintertime uncertainty in the northern hemisphere. The Northern Atlantic shift seasonally to wind-driven uncertainty drives the elevated boreal wintertime uncertainty averages, leading to the interesting global uncertainty correlations regardless of the opposite seasons. Year-round, polar regions are dominated by the gas exchange transfer velocity uncertainty, whereas the subtropics are driven by wind-driven uncertainty. pCO₂driven uncertainty dominates in the non-polar regions with lower observational availability, such as the southern Indian Ocean and the southern hemisphere open Pacific ocean.

While we include the main uncertainty and bias terms in our calculation that have been identified in recent literature (Gloege et al., 2021; Hauck, Nissen, et al., 2023; Roobaert et al., 2018), there are additional sources of

Uncertainty Sources by Percentage of Influence on Total Uncertainty					
		January	July		
Total	k	36%	32%		
	pCO ₂	25%	28%		
	wind	39%	40%		
Polar Region	k	50%	46%		
	pCO ₂	25%	24%		
	wind	25%	30%		
30° latitudinal region	k	24%	23%		
	pCO ₂	27%	30%		
	wind	49%	47%		

19448007, 2024, 4, Downloaded from http: wiley.com/doi/10.1029/2023GL106636 by Ifremer Centre Bretagne Blp, Wiley Online ibrary on [06/03/2025]. See the T Q.A

Acknowledgments

The Surface Ocean CO₂ Atlas (SOCAT) is an international effort, endorsed by the International Ocean Carbon Coordination Project (IOCCP), the Surface Ocean Lower Atmosphere Study (SOLAS) and the Integrated Marine Biosphere Research (IMBeR) program, to deliver a uniformly quality-controlled surface ocean CO. database. The many researchers and funding agencies responsible for the collection of data and quality control are thanked for their contributions to SOCAT. The research is supported by EUREC4A-OA JPI Ocean funded by the Federal Ministry of Education and Research (BMBF). PL acknowledges support from the VLIZ ICOS carbon data collection work from Research Foundation Flanders (FWO) contract I001821N. AJ acknowledges additional support from NASA GMAO and ESSIC at the University of Maryland. We thank J. Marotzke and D. Nielsen from the Max Planck Institute for Meteorology in Hamburg for their comments.Open Access funding enabled and organized by Projekt DEAL.

uncertainty that are not considered here, for example, related to coastal systems, varying river input, the cool skin effect (Takahashi et al., 2009; Watson et al., 2020) or the optimal ensemble size for model subsampling. Most of these are either still under debate or cannot be resolved spatially and temporally at present. Furthermore, measurement uncertainty is not included, as Landschützer et al., 2014 shows that this term is small compared to the much larger uncertainty from extrapolating sparse measurements.

Our study highlights future areas where process research is needed to reduce uncertainty and biases, for example, the quantification of the influence of sea ice. We demonstrate the need to continue improving parameterizations and understanding the impact of variability in carbon flux estimations in order to improve our ability to quantify marine carbon budgets.

Data Availability Statement

Results from the neural-network based SOM-FFM method and uncertainty quantification are publicly available online (Jersild et al., 2023). The source code for the method can be found at (Jersild, 2023). The MPIM-HAMOCC model output can be found (Hauck, Landschützer, et al., 2023).

References

- Bakker, D., Pfeil, H., Landa, C., Metzl, N., O'Brien, K., Olsen, A., et. al. (2016). A mult-decade record of high-quality fCO₂ data in version 3 of the surface ocean CO₂ atlas (SOCAT). *Earth System Science Data*, 8.
- Dickson, A. G., Sabine, C. L., & Christian, J. R. (2007). Guide to best practices for ocean CO₂ measurements, IOCCP Report 8 (Vol. 3). PICES Special Publication.
- Dlugokencky, E., Thoning, K., Lan, X., & Tans, P. (2021). Noaa greenhouse gas reference from atmospheric carbon dioxide dry air mole fractions from the NOAA GML carbon cycle cooperative global air sampling network.
- Fay, A. R., Gregor, L., Landschützer, P., McKinley, G. A., Gruber, N., Gehlen, M., et al. (2021). Seaflux: Harmonization of air-sea CO₂ fluxes from surface pCO₂ data products using a standardized approach. *Earth System Science Data*, 13(10), 4693–4710. https://doi.org/10.5194/essd-13-4693-2021
- Fay, A. R., & McKinley, G. (2013). Global trends in surface ocean pCO₂ from in situ data. Global Biogeochemical Cycles, 27(2), 541–557. https:// doi.org/10.1002/gbc.20051
- Fay, A. R., & McKinley, G. A. (2014). Global ocean biomes: Mean and time-varying maps (NetCDF 7.8 mb). https://doi.org/10.1594/ PANGAEA.828650
- Friedlingstein, P., O'Sullivan, M., Jones, M. W., Andrew, R. M., Bakker, D. C., Hauck, J., et al. (2023). Global carbon budget 2023. Earth System Science Data, 15(12), 5301–5369. https://doi.org/10.5194/essd-15-5301-2023
- Garbe, C., Rutgersson, A., Boutin, J., de Leeuw, G., Delille, B., Fairall, C., et al. (2014). Ocean-atmosphere interactions of gases and particles. Springer Earth System Sciences.
- Gloege, L., McKinley, G. A., Landschützer, P., Fay, A., Frölicher, T. L., Fyfe, J. C., et al. (2021). Quantifying errors in observationally based estimates of ocean carbon sink variability. *Global Biogeochemical Cycles*, 35(4), e2020GB006788. https://doi.org/10.1029/2020gb006788

Good, S., Martin, M., & Rayner, N. (2013). En4: Quality controlled ocean temperature and salinity rofiles and analyses with uncertainty estimates version 4.2.2. *Journal of Geophysical Research: Oceans, 118*(12), 6704–6716. https://doi.org/10.1002/2013jc009067

Gruber, N., Bakker, D. C., DeVries, T., Gregor, L., Hauck, J., Landschützer, P., et al. (2023). Trends and variability in the ocean carbon sink. *Nature Reviews Earth & Environment*, 4(2), 119–134. https://doi.org/10.1038/s43017-022-00381-x

- Hauck, J., Landschützer, P., Mayot, N., & Jersild, A. (2023). Global carbon budget 2023, surface ocean fugacity of CO₂ (fCO₂) and air-sea CO₂ flux of individual global ocean biogeochemical models and surface ocean fCO₂-based data-products [Dataset]. Zenodo. https://doi.org/10. 5281/zenodo.10222484
- Hauck, J., Nissen, C., Landschützer, P., Rödenbeck, C., Bushinsky, S., & Olsen, A. (2023). Sparse observations induce large biases in estimates of the global ocean CO₂ sink: An ocean modeling subsampling experiment (p. 381). Philosophical Transactions: the Royal Society Publishing. Hauck, J., Zeising, M., Quére, C. L., Gruber, N., Bakker, D. C., Bopp, L., et al. (2020). Consistency and challenges in the ocean carbon sink
- Hatter, J., Zeinne, M., Quert, C. L., Ondor, N., Darker, D. C., Dopp, L., et al. (2020). Constantly and the ocean carbon sink estimate for the global carbon budget. Frontiers in Marine Science, 7. https://doi.org/10.3389/fmars.2020.571720 Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. Quarterly Journal
- fersbach, H., Bell, B., Bernstord, P., Hirahara, S., Horanyi, A., Munoz-sabater, J., et al. (2020). The EKAS global reanalysis. *Quarterly Journal* of the Royal Meteorological Society, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Ho, D., Law, C., Smith, M., Schlosser, P., Harvey, M., & Hill, P. (2006). Measurements of air-sea gas exchange at high wind speeds in the southern ocean: Implications for global parameterizations. *Geophysical Research Letters*, 33(16), L16611. https://doi.org/10.1029/ 2006gl026817
- Huang, B., Liu, C., Banzon, V., Freeman, E., Graham, G., Hankins, B., et al. (2020). Improvements of the daily optimum interpolation sea surface temperature (doisst) version 2.1. *Journal of Climate*, 34(8), 2923–2939. https://doi.org/10.1175/jcli-d-20-0166.1
- Ilyina, T., Six, K. D., Segschneider, J., Maier-Reimer, E., Li, H., & Núñez-Riboni, I. (2013). Global ocean biogeochemistry model HAMOCC: Model architecture and performance as component of the mpi-earth system model in different CMIP5 experimental realizations. *Journal of Advances in Modeling Earth Systems*, 5(2), 287–315. https://doi.org/10.1029/2012ms000178
- Jersild, A. (2023). SOMFFN (version 2022) [Software]. Github. Retrieved from https://github.com/aljersild/SOMFFNv2022
- Jersild, A., Landschützer, P., Gruber, P., & Bakker, D. (2023). An observation-based monthly global gridded sea surface pco2 and air-sea CO₂ flux product from 1982 onward its monthly climatology (ncei accession 0160558) [Dataset]. National Oceanic and Atmospheric Administration. Retrieved from https://www.ncei.noaa.gov/access/ocean-carbon-acidification-data-system/oceans/SPCO2_1982_present_ETH_SOM_FFN. html
- Kanamitsu, M., Ebisuzaki, W., Woollen, S.-K., Yang, J., Hnilo, J., Fiorino, M., & Potter, G. (2002). Ncep-doe amip-ii reanalysis (r-2). Bulletin of the American Meteorological Society, 83(11), 1631–1644. https://doi.org/10.1175/bams-83-11-1631(2002)083<1631:nar>2.3.co;2

by the



- Kanlay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L., et al. (1996). The NCEP/NCAR 40-year reanalysis project. Bulletin of the American Meteorological Society, 77(3), 437–472.
- Landschützer, P., Gruber, N., & Bakker, D. (2016). Decadal variations and trends of the global ocean carbon sink. *Global Biogeochemical Cycles*, 30(10), 1396–1417. https://doi.org/10.1002/2015gb005359
- Landschützer, P., Gruber, N., Bakker, D., & Schuster, U. (2014). Recent variability of the global ocean carbon sink. Global Biogeochemical Cycles, 28(9), 927–949. https://doi.org/10.1002/2014gb004853
- Landschützer, P., Gruber, N., Bakker, D., Schuster, U., Nakoaka, S., Payne, M., et al. (2013). A neural network-based estimate of the seasonal to inter-annual variability of the Atlantic Ocean carbon sink. *Biogeosciences*, 10(11), 7793–7815. https://doi.org/10.5194/bg-10-7793-2013
- Landschützer, P., Tanhua, J., nad Behncke, T., & Keppler, L. (2023). Sailing through the southern seas of air-sea CO₂ flux uncertainty. The Royal Society.
- McKinley, G. A., Bennington, V., Meinshausen, M., & Nicholls, Z. (2023). Modern air-sea flux distributions reduce uncertainty in the future ocean carbon sink. *Environmental Research Letters*, 18(4), 044011. https://doi.org/10.1088/1748-9326/acc195
- Montégut, C. D. B., Madec, G., Fischer, A. S., Lazar, A., & Iudicone, D. (2004). Mixed layer depth over the global ocean: An examination of profile data and a profile-based climatology. *Journal of Geophysical Research*, 109(C12), C12003. https://doi.org/10.1029/2004jc002378
- Naegler, T. (2009). Reconciliation of excess 14c-constrained global co2 piston velocity estimates. *Tellus B: Chemical and Physical Meteorology*, 61(2), 372–384. https://doi.org/10.3402/tellusb.v61i2.16836
- Peterson, P. H., Stöckl, D., Westgard, J. O., Sandberg, S., Linnet, K., & Thienpont, L. (2001). Models for combining random and systematic errors: Assumptions and consequences for different models. *Clinical Chemistry and Laboratory Medicine*.
- Randerson, J., Lindsay, K., Munoz, E., Fu, W., Moore, J., Hoffmann, F., et al. (2015). Multicentury changes in ocean and land contributions to the climate-carbon feedback. *Global Biogeochemical Cycles*, 29(6), 744–759. https://doi.org/10.1002/2014gb005079
- Rayner, N., Parker, D., Horton, E., Folland, C., Alexander, L., Rowell, D., et al. (2003). Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century. *Journal of Geophysical Research*, 108(D14), 4407. https://doi.org/10.1029/ 2002jd002670
- Ridge, S. M., & McKinley, G. A. (2021). Ocean carbon uptake under aggressive emission mitigation. *Biogeosciences*, 18(8), 2711–2725. https:// doi.org/10.5194/bg-18-2711-2021
- Rödenbeck, C., Bakker, D., Gruber, N., lida, Y., Jacobson, A., Jones, S., et al. (2015). Data-based estimates of the ocean carbons ink variability— First results of the surface ocean pCO₂ mapping intercomparison (SOCOM). *Biogeosciences*, 12(23), 7251–7278. https://doi.org/10.5194/bg-12-7251-2015
- Roobaert, A., Laruelle, G., Landschützer, P., Gruber, N., Chou, L., & Regnier, P. (2019). The spatiotemporal dynamics of the sources and sinks of CO₂ in the global coastal ocean. *Global Biogeochemical Cycles*, *33*(12), 1693–1714. https://doi.org/10.1029/2019gb006239
- Roobaert, A., Laruelle, G., Landschützer, P., & Regnier, P. (2018). Uncertainty in the global oceanic CO₂ uptake induced by wind forcing: Quantification and spatial analysis. *Biogeosciences*, 15(6), 1701–1720. https://doi.org/10.5194/bg-15-1701-2018
- Sarmiento, J. L., & Gruber, N. (2006). Ocean biogeochemical cycles. Princeton University Press. https://doi.org/10.2307/j.ctt3fgxqx
- Sweeney, C., Gloor, E., Jacobson, A. R., Key, R. M., McKinley, G., Sarmiento, J. L., & Wanninkhof, R. (2007). Constraining global air-sea gas exchange for CO₂ with recent bomb 14c measurements. *Global Biogeochemical Cycles*, 21(2), GB2015. https://doi.org/10.1029/ 2006gb002784
- Takahashi, T., Sutherland, S. C., Wanninkhof, R., Sweeney, C., Feely, R. A., Chipman, D. W., et al. (2009). Climatological mean and decadal change in surface ocean pCO₂, and net sea-air CO₂ flux over the global oceans. *Deep Sea Research Part II: Topical Studies in Oceanography*, 56(8), 554–577. https://doi.org/10.1016/j.dsr2.2008.12.009
- Wanninkhof, R. (2014). Relationship between wind speed and gas exchange over the ocean revisited. *Limnology and Oceanography: Methods*, 12(6), 351–362. https://doi.org/10.4319/lom.2014.12.351
- Wanninkhof, R. (1992). Relation between wind speed and gas exchange over the ocean. *Journal of Geophysical Research*, *97*, 7373–7383. Watson, A. J., Schuster, U., Shutler, J. D., Holding, T., Ashton, I. G., Landschützer, P., et al. (2020). Revised estimates of ocean-atmosphere
- Watson, A. J., Schuster, U., Shutler, J. D., Holding, T., Ashton, I. O., Landschutzer, P., et al. (2020). Revised estimates of ocean-annosphere CO₂ flux are consistent with ocean carbon inventory. *Nature Communications*, 11(1), 4422. https://doi.org/10.1038/s41467-020-18203-3
 Woolf, D., Shutler, J., Goddijn-Murphy, L., Watson, A., Capron, B., Nightingale, P., et al. (2019). Key uncertainties in the recent air-sea flux of
- Woolf, D., Shutler, J., Goddijn-Murphy, L., Watson, A., Capron, B., Nightingale, P., et al. (2019). Key uncertainties in the recent air-sea flux of CO₂. Global Biogeochemical Cycles, 33, 1563. https://doi.org/10.1029/2018GB006041

References From the Supporting Information

Butterworth, B. J., & Miller, S. D. (2016). Air-sea exchange of carbon dioxide in the southern ocean and Antarctic marginal ice zone. *Geophysical Research Letters*, 43(13), 7223–7230. https://doi.org/10.1002/2016gl069581

Illowsky, B., & Dean, S. (2013). Introductory statistics, chap. 6. OpenStax.